\documentstyle{amsppt}
\magnification=1200
\define\nin{\noindent}
\loadmsam
\loadbold
\loadmsbm
\loadeufm
\loadeusm
\UseAMSsymbols
\NoBlackBoxes
\font\boldtitlefont=cmb10 scaled\magstep2
\font\sectionfont=cmb10 scaled\magstep1
\NoRunningHeads
\noindent
\centerline{\bf Math 226.02/226T.01}
\centerline{\bf Quiz 7}
\centerline{\bf October 30, 1997}
\vskip 1. in
\noindent
{\bf Name:}\hskip .5 in \hrule width5cm \hfil

\bigskip
\noindent
Determine the Taylor series about the point $x_0 = 2$ for the
function $ f(x) = \frac{4}{3+x}$.  Find also the radius of
convergence of the series.


\vskip4. in
\noindent
{\bf Answer:} \hrule width5cm \hfil
\bigskip
\noindent
Sign the pledge "On my honor, I have neither given nor received
unauthorized aid on this quiz."
\bigskip
\noindent
{\bf Signature:} \hrule width5cm \hfil


 \end